Author's response to reviews

Title: Does the world need a scientific society for research on how to improve healthcare?

Authors:

Michel Wensing (M.Wensing@iq.umcn.nl)
Jeremy M Grimshaw (jgrimshaw@ohri.ca)
Martin P Eccles (martin.eccles@newcastle.ac.uk)

Version: 2 Date: 13 January 2012

Author's response to reviews:

dear Martin,

I have included the comments of the contributors and the URL link in this revised manuscript.